

Diverging patterns of introgression from Schistosoma bovis across S. haematobium African lineages

Olivier REY, Eve Toulza, Jérôme Boissier, Cristian Chaparro, Jean-François Allienne, Julien Kincaid-Smith, Eglantine Mathieu-Bégné, Fiona Allan, David Rollinson, Bonnie Webster

▶ To cite this version:

Olivier REY, Eve Toulza, Jérôme Boissier, Cristian Chaparro, Jean-François Allienne, et al.. Diverging patterns of introgression from Schistosoma bovis across S. haematobium African lineages. PLoS Pathogens, 2021, 17 (2), pp.e1009313. 10.1371/journal.ppat.1009313 . hal-03176500

HAL Id: hal-03176500 https://hal.science/hal-03176500v1

Submitted on 22 Mar 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés. Check for updates

Citation: Rey O, Toulza E, Chaparro C, Allienne J-F, Kincaid-Smith J, Mathieu-Begné E, et al. (2021) Diverging patterns of introgression from *Schistosoma bovis* across *S. haematobium* African lineages. PLoS Pathog 17(2): e1009313. https:// doi.org/10.1371/journal.ppat.1009313

Editor: Michelle Steinauer, Western University of Health Sciences College of Osteopathic Medicine of the Pacific NW, UNITED STATES

Received: June 18, 2020

Accepted: January 13, 2021

Published: February 5, 2021

Copyright: © 2021 Rey et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The datasets were deposited on the GenBank SRA repository. The raw genomic datasets are available on GenBank Bioproject PRJNA694548.

Funding: This work was supported by a "Research Quality Grant" from the University of Perpignan (awarded to OR) and by the program HySWARM (ANR-18-CE35-0001) ANR (awarded to JB). The Schistosomiasis Collection at the Natural History Museum (SCAN) is part funded by the Wellcome **RESEARCH ARTICLE**

Diverging patterns of introgression from *Schistosoma bovis* across *S. haematobium* African lineages

Olivier Rey^{1*}, Eve Toulza¹, Cristian Chaparro¹, Jean-François Allienne¹, Julien Kincaid-Smith^{1,2}, Eglantine Mathieu-Begné¹, Fiona Allan^{3,4}, David Rollinson^{3,4}, Bonnie L. Webster^{3,4}, Jérôme Boissier¹

 Univ. Montpellier, CNRS, IFREMER, UPVD, IHPE, Perpignan, France, 2 Centre for Emerging, Endemic and Exotic Diseases (CEEED), Department of Pathobiology and Population Sciences (PPS), Royal Veterinary College, University of London, Hawkshead Campus, Herts, United Kingdom, 3 Wolfson Wellcome Biomedical Laboratories, Department of Life Sciences, Natural History Museum, London, United Kingdom, 4 London Centre for Neglected Tropical Disease Research, Imperial College London School of Public Health, London, United Kingdom

* olivier.rey@univ-perp.fr

Abstract

Hybridization is a fascinating evolutionary phenomenon that raises the question of how species maintain their integrity. Inter-species hybridization occurs between certain Schistosoma species that can cause important public health and veterinary issues. In particular hybrids between Schistosoma haematobium and S. bovis associated with humans and animals respectively are frequently identified in Africa. Recent genomic evidence indicates that some S. haematobium populations show signatures of genomic introgression from S. bovis. Here, we conducted a genomic comparative study and investigated the genomic relationships between S. haematobium, S. bovis and their hybrids using 19 isolates originating from a wide geographical range over Africa, including samples initially classified as S. haematobium (n = 11), S. bovis (n = 6) and S. haematobium x S. bovis hybrids (n = 2). Based on a whole genomic sequencing approach, we developed 56,181 SNPs that allowed a clear differentiation of S. bovis isolates from a genomic cluster including all S. haematobium isolates and a natural S. haematobium-bovis hybrid. All the isolates from the S. haematobium cluster except the isolate from Madagascar harbored signatures of genomic introgression from S. bovis. Isolates from Corsica, Mali and Egypt harbored the S. bovis-like Invadolysin gene, an introgressed tract that has been previously detected in some introgressed S. haematobium populations from Niger. Together our results highlight the fact that introgression from S. bovis is widespread across S. haematobium and that the observed introgression is unidirectional.

Author summary

Hybridization is a fascinating evolutionary phenomenon that raises the question of how species maintain their integrity. Inter-species hybridization occurs between certain

Trust (grant 104958/Z/14/Z). EMB received a salary from the ANR (ANR-18-CE35-0001). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Schistosoma species that can cause important public health and veterinary issues. In particular hybrids between *Schistosoma haematobium* and *S. bovis* associated with humans and animals respectively are frequently identified in Africa. Recent genomic evidence indicates that some *S. haematobium* populations show signatures of genomic introgression from *S. bovis*. Here we conducted a comparative genomic study to assess the genomic diversity within *S. haematobium* and *S. bovis* species and genetic differentation at the genome scale between these two sister species over the African continent. We also investigated traces of possible ancient introgression from one species to another. We found that *S. haematobium* display low genetic diversity compared to *S. bovis*. We also found that most *S. haematobium* samples harbor signature of past introgression with *S. bovis* at some genomic positions. Our results strongly suggest that introgression occurred long time ago and that such introgression is unidirectional from *S. bovis* within *S. haematobium*. Such introgression event(s) result in diverging patterns of genomic introgression across *S. haematobium* lineages.

Introduction

Hybridization is ubiquitous within the tree of life that raises the question of how species maintain their integrity [1-3]. Although hybridization may often result in the production of nonviable or non-fertile offspring, increasing evidence suggests that certain interspecific crosses produce not only viable progeny, but those progeny may be more successful than their parental species, at least during the early generations (e.g. hybrid vigor) [4,5]). For longer evolutionary timescales, genomic introgression resulting from hybridization events may be adaptive when genes and/or alleles from one species are maintained through subsequent generations and backcrosses by natural selection, becoming part of the genomic background [4,6,7]. One classical example of adaptive introgression relates to our own modern human lineage that has acquired genes involved in important functions such as immunity, pigmentation or resistance to hypoxia, originating from at least two introgression events with related ancestral lineages including Neanderthals and Denisovans [6]. It is thus now generally acknowledged that hybridization constitutes an important fuel for evolution and adaptation [8].

Hybridization also occurs between parasites and might have dramatic evolutionary outcomes [9]. In particular, several studies suggest that interspecific crosses among parasite species can generate lineages that display increased virulence compared to parental species, possibly leading to atypical pathologies and/or a wider host spectrum [9–11] and hence new epidemiological dynamics [9,12]. Hybridization and introgression events in parasites are thus likely to pose serious challenges for prevention, control and therapies against parasitic diseases [9].

Schistosomes are gonochoric parasitic flatworms with two successive hosts, a vertebrate as the definitive host in which adult worms reproduce sexually, and a freshwater snail species in which parasites undergo clonal multiplication. *Schistosoma spp.* can cause chronic and acute intestinal or urogenital schistosomiasis posing serious public health and veterinary issues. In tropical and subtropical regions it is thought that 143 million people are infected with schistosomes, with an estimated 24 thousand people dying each year from related pathology (IHME; Global Burden Disease). The *Schistosoma* genus diverged approximately 22 million years (my) ago and is comprised of 23 known species that specifically infect a broad range of mammals including Carnivora, Cetartiodactyla, Perissodactyla, Rodentia and Primates, including humans, as definitive hosts [13,14]. Inter-species hybridization is known to occur between certain *Schistosoma* species, with experimental inter-species crosses demonstrating important variation in the resulting offsprings's fitness dependent on the species involved [15]. Natural

inter-species hybrids have also been recognized between closely related sister species (e.g. S. haematobium and S. guineensis) [16], but also between more distantly related species that diverged up to ~12.5 my ago (e.g. between S. mansoni and S. haematobium) [17], that reside in the same definitive host. However, the evolutionary distance of the species involved does dictate hybrid viability. Hybrids resulting from closely related species that have different mammalian host preferences (e.g. non-human animals versus humans) are generating much research attention due to potential zoonotic consequences. For instance, S. bovis infecting ungulates, especially livestock, and S. haematobium infecting humans diverged 3.3 my ago but can successfully interbreed as testified by empirical evidence from the field [18-20], and by experimental studies [18,21]. These hybrids between animal and human associated Schistosoma species have recently received considerable attention due to their ease of inter-breeding, geographically widespread sympatric transmission, high-prevalences and moreover due to the potential zoonotic implications that this hybridization system suggests. Additionally, host pathology cannot be disregarded and in this context, J. Kincaid-Smith [22] experimentally demonstrated hybrid vigor in the first generation hybrids between a S. bovis and S. haematobium laboratory cross. The hybrids were compatible with natural snail hosts of both parental species, (i.e. Bulinus truncatus and Planorbarius metidiensis) and were more virulent than the parental strains within the vertebrate definitive host (in this case the experimental rodent Mesocricetus auratus). Additionally, this hybrid lineage was found to have established itself outside its native tropical range; in Corsica (France), where natural populations of B. truncatus occur. The hybrids established in Corsica display a highly introgressed genome, 77% was of S. haematobium origin, and 23% of S. bovis origin [21]. In this context, it was also hypothesised that this introgression may have enabled the establishment of urogenital transmission outside of its typical endemic zones [19].

A recent genomic study showed that *S. haematobium* populations from Niger were introgressed with a small proportion of *S. bovis* genomic background (i.e. ~3–8%) [23]. The authors suggest that this likely results from ancient hybridization and that the resulting introgression might have been adaptive with strong directional selection of a specific adaptive gene (i.e. *Invadolysin*) related to host-pathogen interactions originating from *S.bovis* and becoming fixed in these introgressed populations. Interestingly, this introgressed adaptive gene was also identified in the reference genome of *S. haematobium*, originally sampled in Egypt and now reared as a laboratory strain for over half a century [23,24]. These recent findings challenge the genetic integrity of the species *S. haematobium* and call into question whether pure *S. haematobium* lineages actually persist in Africa. Further genomic analyses are thus required to best characterize *S. haematobium* populations and unravel the genomic relationships between *S. haematobium*, *S. bovis* and their hybrids across all endemic countries.

In this context, we performed genome-wide analyses based on *Schistosoma* samples originating from a wide range of East and West African countries and initially characterized as *S. haematobium*, *S. bovis* or *S. haematobium*—*S. bovis* hybrids as well as an experimentally generated *S. haematobium-bovis* F1 hybrid (Fig 1; Table 1). Specifically, we developed a set of genomewide SNPs from whole genome sequencing of 19 isolates to study the genomic relationship between the two parental species *S. haematobium* and *S. bovis* and their hybrids. Based on these SNPs we investigated possible traces of introgression events between these two sister species.

Material and methods

Sampling collection

The sampling strategy consisted of sampling individual adult male worms which had been isolated from a range of African countries, to capture as much genomic diversity as possible in





https://doi.org/10.1371/journal.ppat.1009313.g001

both what we consider to be *S. haematobium* and *S. bovis*. To accomplish this aim, we combined samples recently collected in the field (N = 4), historical samples from the SCAN project (N = 12) [25] and samples from laboratory strains (N = 3) including one *S. haematobium* sample from Egypt and one *S. bovis* sample from Spain, both lineages having been passaged in the laboratory for more than 30 years. The Spanish *S. bovis* isolate was used for assembling the

Species affiliation	Country	Collection year	Live stage at collection	Host retrieved from
S. bovis (S.b)	Kenya	1985	Cercariae	B. tropicus
	Nigeria	1988	Cercariae	B. truncatus
	Senegal	2000	Miracidia	Cow
	Uganda	2006	Cercariae	Undefined snail
	Ethiopia	2018	Adult	Cow
	Spain	Lab.strain*	Adult	Hamster
S. haematobium (S.h)	Egypt	Lab.strain*	Adult	Hamster
	Cameroon	2001	Miracidia	Human
	Gambia	1987	Cercariae	Undefined snail
	Guinea-Bissau	1990	Cercariae	B. senegalensis
	Madagascar	1987	Miracidia	Human
	Nigeria	1986	Cercariae	B. truncatus
	Sudan	1984	Miracidia	Human
	Mali	1994	Cercariae	Undefined snail
	Tanzania (Zanzibar)	1994	Miracidia	Human
	IvoryCoast	2018	Miracidia	Human
	Senegal	2019	Miracidia	Human
S.b—S.h hybrid	F1 Hybrid	Lab.strain**	Adult	Hamster
	Corsica	2013	Miracidia	Human

Table 1. Sample information. Samples from countries in bold were obtained from SCAN [25].

* The laboratory strains from Spain and Egypt were maintained in the culturing facilities at the University of Perpignan, previously provided by Ana Oleaga [54] and the Biomedical Research Institute, Rockville, Maryland [55], respectively.

** The F1 hybrid was generated in 2015 by crossing a female S. bovis from Spain and male S. haematobium from Cameroon (Barombi-Kotto) [21].

https://doi.org/10.1371/journal.ppat.1009313.t001

reference genome of *S. bovis*. The remaining laboratory strain consisted of a *S. haematobium-bovis* F1 hybrid resulting from the cross between a female *S. bovis* (Spain) and a male *S. haematobium* (Cameroon) generated at the University of Perpignan animal facility [21]. Our final sample set consisted of 19 adult males that were either collected as adults or originally as free-living stages (Table 1). If isolated from larval stages, adult worms were obtained from laboratory life-cycles using hamsters (*Mesocricetus auratus*) for *S. haematobium* or mice (*Mus musculus*) for *S. bovis* as the definitive hosts.

Genomic DNA extraction

Total gDNA was extracted from each individual using the Qiamp DNA Micro Kit (Qiagen), following the protocol for "isolation of genomic DNA from tissues" and ultimately eluted in 25 μ l of distilled water. The concentration and quality of each individual gDNA extract was assessed using a Qubit 3.0 fluorometer (Invitrogen).

Library preparation and genome-wide sequencing process

The supposed nature of each isolate (i.e. *S. haematobium*; *S. bovis* or hybrid) was first validated based on a classical genotyping at the mitochondrial *COI* and at the *ITS2* nuclear gene [18,26,27]. Individual genomic libraries were prepared using the Nextera XT DNA Library Prep Kit (Illumina). One nanogram of genomic DNA was used as the template for each individual sample. Libraries were cleaned using 90 μ L of AMPure XP beads (Beckman Coulter) to enrich libraries with fragments ranging from 300 to 500 bp. The quality of each library was then assessed using Agilent Technology 2011 Bioanalyser with a High Sensitivity DNA chip as

recommended in the protocol. Individual libraries were sent to the Bio-Environment platform (University of Perpignan) where they were normalized and pooled before being paired-end sequenced within two independent runs (2x150 bp) on a NextSeq 550 (Illumina).

Development of genome-wide SNPs

Demultiplexed FASTQ R1 and R2 files were uploaded to the Galaxy web platform [28]. R1 and R2 read datasets for each sample were first compiled into a single list of dataset pairs. Raw data quality within each list was assessed using the FastQC tool [29]. Reads were then trimmed to remove low quality sequences and the sequencing and index adapters. Once cleaned, reads were aligned to the *S. haematobium* (Egyptian strain) reference genome ShV2_May19_020001-666.fa (of 370 Mb and which contains 666 scaffolds; available on Gen-Bank BioProject PRJNA78265) using bowtie2 [30] using default parameters settings. Aligned reads from each sample were merged together in a unique.BAM file and alignment statistics were retrieved using the Samtools flagstat tool [31]. The total numbers of reads before and after preprocessing together with individual alignment rate are provided in <u>S1 Table</u>.

SNP calling was performed using GATK 4.0. [32]. Within GATK the HaplotypeCaller tool was first used on individual.BAM files for each sample. The resulting individual ".gvcf" files were next combined into a unique ".gvcf" file using the CombineGVCFs tool. SNPs were then filtered using the VariantFiltration tool and setting the following filtering parameters: FS > 45.0; QD < 2.0; MQ < 40.0; MQRankSum < -12.5 and ReadPosRankSum <-8.0. Finally, the genotyping file was obtained by filtering genotypes using vcftools version 0.1.16 [33] based on the former filtering parameters (—remove-filtered-all), removing indels (—remove-indels), prohibiting any missing data (—max-missing 1), keeping SNPs with minimal depth of 8 (—minDP 8) and minor alleles present in at least 3 copies (—mac 3). Finally, from the initial 57,615 SNPs obtained from GATK, only biallelic sites were conserved and all SNPs belonging to the mitochondrial genome (i.e. "020257" in the genome of reference) were removed to specifically focus on the nuclear genome. These filtering steps resulted in a set of 56,181 SNPs that were used for subsequent analyses.

Genomic analyses

Clustering analysis. To analyse the global structure of genetic variation, a principal component analysis (PCA) was performed based on the 56,181 filtered SNPs using the dudi.pca command from the "ade4" package [34] implemented in R version 4.0.2 [35]. Isolates were plotted according to their coordinates on the first two PCA axes using the "factoextra" package in R [36]. To detect possible finer genetic structure among isolates, three additional PCA plots were performed to represent the spatial distribution of each isolates within three other dimensional spaces including the two following PCA axes (i.e. Axes 2 and 3; Axes 2 and 4 and Axes 3 and 4).

Within and between species genomic diversity. Based on the 56,181 initial filtered SNPs, individual observed heterozygosity (Hobs) was first calculated for each isolate. Difference in mean Hobs between species was assessed using a Mann-Whitney test in R. According to the results from the clustering analysis (see results), the six nominal *S.bovis* isolates were assigned to a *S. bovis* cluster, and the 11 nominal *S. haematobium* isolates together with the hybrid isolate from Corsica were assigned to a *S. haematobium* cluster. The same assignment of isolates among the two defined clusters was set and the F1 hybrid generated in the laboratory was excluded from all subsequent analyses. Second, the average pairwise nucleotide difference among isolates (Π) was computed within each cluster and pairwise nucleotide differences computed at each site (N = 56,181) were compared between the *S. bovis* and *S. haematobium* clusters, using a non-parametric paired Wilcoxon test in R. Third, Tajima's Ds were estimated

for each species. To statistically compare Tajima's D between species, a series of Tajimas'Ds was computed along non-overlapping 50-kb sliding windows overall genomic scaffolds within each species. Windows that contained less than eight SNPs were discarded and only the 50-kb sliding windows in which Tajima's Ds could be estimated for *S. bovis* and *S. haematobium* (i.e. with at least one segregating site within each species) were kept (N = 4035). The series of estimated Tajima's Ds obtained for each species were next compared using a paired Wilcoxon test. Finally, overall absolute genomic divergence (Dxy) between *S. bovis* and *S. haematobium* was estimated. All nucleotide diversity and divergence estimates were computed using the 'PopGenome' R Package [37].

Spatial genomic structure within the *S. bovis* **and** *S. haematobium* **clusters.** To investigate any potential spatial genetic structure among isolates within each species, we computed pairwise Nei's genetic distances matrices [38], between each isolate within each species (i.e. *S. haematobium* = all 11 nominal *S. haematobium* isolates + the Corsican hybrid isolate; and *S. bovis* = the six nominal *S. bovis* isolates), based on the initial filtered 56,181 SNPs and using the dartR package [39]. Correlation between the resulting matrices of pairwise Nei's distances and pairwise geographical distances computed between the geographical origin of each isolate were next tested within each species using two independent Mantel tests as implemented in the 'ape' package [40] in R. Significance of correlations were assessed by comparing the obtained Z-scores to 10,000 Z-values generated after bootstrapping. Because uncertainty existed concerning the precise sampling location for some isolates, we considered the centroid of each country of origin obtained from QGIS v. 2.8.12 (QGIS Development Team).

Investigating signatures of introgression between S. bovis and S. haematobium. Introgression events are expected to result in the acquisition of genomic tracts from one species to another that can persist -and thus be detectable- within the introgressed lineage over time. This is dependent on time since the introgression event, the genomic context (e.g. genomic locality) of the foreign DNA tracts in the introgressed genome, the recombination rate, the selective advantage of the introgressed DNA tracts [1,3]. Using a genomic sliding windows approach, we thus investigated the presence of genomic regions that display significantly decreased divergence between each isolate and all isolates from its sister species (i.e. one versus all). Thus, to detect possible genomic tracts that could have been introgressed from S. bovis in each of the S. haematobium isolates, we computed Dxy estimates between each S. haematobium isolate (including the hybrid Corsican lineage) and all the S. bovis isolates along 50-kbs windows every 10 kbs overall genomic scaffolds. The same genomic windows were used to compute Dxy estimates between each S. bovis isolate and all the S. haematobium isolates (including the hybrid Corsican lineage) to detect potential traces of introgression from S. haematobium in each of the S. bovis isolates. The Dxy computations along the 50-kbs sliding windows were performed using the R package 'PopGenome' [37]. All 50 kbs-windows that contained less than eight SNPs were discarded for subsequent analyses. It was considered that a sliding window displayed significantly decreased divergence when the Dxy value computed was intrinsically lower than the lowest values (0.1%) obtained overall the windows (with > eight SNPs) and overall the isolates (N = 239,076; i.e. 13,282 windows x 18 isolates). This 0.1% threshold corresponded to a Dxy value of 0.102.

Along each scaffold, unique genomic tracts were reconstructed from all overlapping sliding windows that displayed significantly decreased divergence according to the above 0.1% threshold including the windows with less than eigth SNPs. The absolute size, the total number of SNPs and individual Dxy values (i.e. Dxy computed between each isolate and all isolates from its sister species) were computed for each genomic tract. Finally, we investigated the presence of potential coding genes within the resulting genomic tracts based on the ShV2_May19_SwissProtAnno.gff3 annotation file [41] and using IGV 2.3.92 [42].

Results

Clustering analysis

Overall, 56,181 biallelic SNPs distributed over 239 of the 665 nuclear genomic scaffolds (i.e. ~ 36%) of the *S. haematobium* reference genome (Egyptian strain) were developed into a highquality SNP dataset. Using this dataset, we could clearly distinguish between isolates categorized as either *S. haematobium* or *S. bovis* based on a principal component analysis (i.e. PCA; Fig 2A). In this regard, the overall absolute genomic divergence between *S. bovis* and *S. haematobium* reaches 0.24. As expected, the laboratory generated first generation *S. haematobium-bovis* hybrid (F1) fell mid way between the two parental strains used for its construction, i.e. *S. bovis*



Fig 2. Genome-wide genetic diversity and genetic differentation within *S. haematobium* and *S. bovis*. **A.** Spatial representation of all isolates in a PCA dimension defined along the two first axis that explain 63.1% and 6.5% of the total genomic variance respectively. Isolates identified as *S. haematobium* are in blue, *S. bovis* isolates are in red, the Corsican hybrid is in yellow and the experimentally bred F1 hybrid is in grey. **B.** Mean individual heterozygosity within *S. bovis* (in red) and *S. haematobium* (including the Corsican lineage; in blue). **C.** Tajimas'D estimates along 4035 50 kbs-sliding windows that contained at least 8 SNPs over 159 scaffolds computed within *S. bovis* (in red) and *S. haematobium* (in blue). **D.** Plot of Nei's distances computed between isolates and geographical distances between their countries of origin (using the countries' centroid as proxy for each locality) within *S. bovis* (in red) and *S. haematobium* (in blue). The genomic structure among *S.bovis* lineages follows a classical significant isolation-by-distance pattern at the African continent scale.

https://doi.org/10.1371/journal.ppat.1009313.g002

from Spain and *S. haematobium* from Cameroon in the PCA two-dimensional space (Fig 2A). No other sample displayed such an intermediate genomic pattern. More surprisingly, according to the first and second PCA axes (respectively representing 63.1% and 6.5% of the overall variance), all *S. haematobium* samples and the hybrid sample from Corsica, grouped into a unique genetic cluster (Fig 2A). The projection of all isolates in the other PCA two-dimensional spaces defined from the third and fourth PCA axes lead to the same pattern with a single condensed cluster with no genomic distinction of the hybrid Corsican isolate from the other *S. haemato-bium* isolates (S1 Fig).

Within and between species genomic diversity

Our results indicate that the *S. haematobium* cluster as defined based on the PCA analysis (i.e. all 11 nominal *S. haematobium* and the Corsican hybrid isolates) harbored less genomic diversity than the *S. bovis* cluster. First, the average level of individual observed heterozygosity (Hobs) was significantly lower in the *S. haematobium* cluster ($H_{obs} = 0.07$) than in the *S. bovis* cluster ($H_{obs} = 0.17$; W = 0; P-value < 0.01; Fig 2B). Additionally, the mean pairwise nucleotide divergence per site is also lower in the *S. haematobium* compared to the *S. bovis* cluster (0.10 *versus* 0.18, V = 566942560; P-value < 0.01). Tajima's D values computed along the genome among the *S. haematobium* isolates (including the Corsican lineage) were lower (although still positive) compared to those computed among the *S. bovis* isolates (0.29 *versus* 0.78; V = 5738688; P-value < 0.01; Fig 2C). Finally, pairwise genetic differentiation estimates between isolates within the two clusters were systematically lower in *S. haematobium* compared to *S. bovis* and while the genetic diversity among *S. bovis* isolates followed an isolation by distance pattern (Z = 7585.7; P-value< 0.01), no such pattern was found among *S. haematobium* isolates (Z = 14809; P-value = 0.15; Fig 2D).

Signature of introgression between S. bovis and S. haematobium

Overall, Dxy values between each isolate and all isolates from its sister species (i.e. one versus all) were computed over 34,852 50-kbs sliding windows along 168 genomic scaffolds. From these initial 34,852 sliding windows, 13,282 contained more than eight SNPs from which 239,076 individual D_{xy} values were retrieved for all the 18 isolates (i.e. all isolates except the F1 laboratory hybrid). These windows covered 53% of the genome (196.51 Mb) and 159 genomic scaffolds. A total of 158 windows distributed over 18 genomic scaffolds harbored significantly decreased genetic divergence between at least one isolate and all isolates combined from the sister species according to our 0.1% threshold. Of these 158 windows, 11 had significant decreases in divergence between at least one S. bovis isolate and all S. haematobium isolates and 147 had significant decreases in divergence between at least one S. haematobium isolate and all S. bovis isolates. Finally, from these sliding windows a total of 29 genomic tracts (i.e. successions of adjacent sliding windows) characterized by extremely low genomic divergence were reconstructed with absolute sizes ranging from 2,456 bp to 630,020 bp and containing from eight to 241 SNPs. Twenty four of these genomic tracts corresponded to genomic regions that display low Dxy values computed between at least one S. haematobium isolate and all S. bovis isolates (Table 2). These genomic tracts were 82,464 bp long on average (ranging from 9,620 bps to 630,020 bps) and contained on average of 23.4 SNPs (from eight to 241, Table 2; Fig 3). The remaining five genomic tracts corresponded to regions that display low Dxy values computed between at least one S. bovis and all the S. haematobium isolates combined (including the Corsican hybrid isolate; Table 3). These five genomic tracts were on average 25,019 bp long (ranging from 2,456 to 48,670 bp) and contained on average of 9.2 SNPs (from eight to 13; Table 3). Finally, over the 24 S. bovis-type genomic tracts found in at least one S. haematobium isolate, 36 genes were identified. In this regard, one genomic tract of length 212,351 bps

Scaffold identity	Number of SNPs	Start position	End position	Tract size (bp)	Cameroon	Corsica	Egypt	Gambia	Guinea Bissau	Ivory Coast	Madagascar	Mali	Nigeria	Senegal	Sudan	Zanzibar
20002	15	86414	113838	27424	0.583	0.061	0.356	0.356	0.139	0.65	0.65	0.433	0.5	0.65	0.65	0.583
20018	54	8423425	8635776	212351	0.145	0.077	0.085	0.549	0.173	0.343	0.59	0.076	0.196	0.373	0.117	0.608
20018	11	8850176	8899446	49270	0.545	0.061	0.303	0.061	0.424	0.485	0.424	0.242	0.303	0.061	0.485	0.424
20040	10	1452306	1486703	34397	0.65	0.083	0.7	0.417	0.7	0.392	0.7	0.75	0.442	0.7	0.7	0.75
20062	13	5240139	5266732	26593	0.308	0.346	0.346	0.308	0.077	0.385	0.346	0.346	0.269	0.077	0.308	0.077
20072	34	1770	136940	135170	0.875	0.488	0.86	0.904	0.904	0.89	0.86	0.071	0.831	0.875	0.743	0.875
20120	8	315078	342710	27632	0.667	0.042	0.604	0.542	0.604	0.667	0.604	0.667	0.542	0.604	0.667	0.604
20120	8	1486323	1522893	36570	0.76	0.094	0.76	0.427	0.146	0.427	0.76	0.76	0.76	0.094	0.76	0.76
20135	241	9647	639667	630020	0.865	0.462	0.872	0.872	0.863	0.847	0.861	0.018	0.805	0.874	0.756	0.859
20135	19	651522	696944	45422	0.798	0.474	0.798	0.825	0.851	0.772	0.772	0.096	0.496	0.851	0.496	0.746
20135	101	813373	1147895	334522	0.77	0.382	0.771	0.743	0.757	0.755	0.724	0.082	0.606	0.767	0.521	0.768
20135	14	1223289	1270364	47075	0.869	0.798	0.798	0.798	0.798	0.798	0.869	0.06	0.833	0.798	0.798	0.94
20135	10	1314098	1323718	9620	0.817	0.817	0.817	0.817	0.817	0.817	0.817	0.083	0.817	0.817	0.817	0.817
20168	29	135	27437	27302	0.483	0.483	0.328	0.31	0.216	0.399	0.483	0.31	0.443	0.086	0.417	0.448
20168	13	1900169	1917247	17078	0.141	0.436	0.09	0.474	0.436	0.122	0.513	0.436	0.436	0.321	0.513	0.436
20173	17	1412975	1494068	81093	0.721	0.25	0.245	0.466	0.779	0.216	0.779	0.407	0.75	0.309	0.093	0.779
20173	8	6405980	6428204	22224	0.073	0.073	0.073	0.51	0.073	0.188	0.552	0.073	0.135	0.51	0.073	0.615
20174	9	1040987	1082699	41712	0.463	0.074	0.269	0.306	0.148	0.63	0.352	0.407	0.574	0.574	0.407	0.352
20174	10	2206173	2244270	38097	0.25	0.4	0.1	0.5	0.167	0.4	0.4	0.25	0.233	0.4	0.25	0.4
20175	11	419051	443512	24461	0.273	0.136	0.136	0.045	0.364	0.091	0.318	0.045	0	0.318	0.136	0.455
20241	20	361697	372646	10949	0.517	0.1	0.492	0.317	0.517	0.542	0.492	0.367	0.517	0.492	0.542	0.517
20241	20	1310224	1358966	48742	0.767	0.429	0.742	0.742	0.742	0.767	0.767	0.092	0.729	0.742	0.704	0.742
20243	12	821454	831590	10136	0.063	0.354	0.479	0.521	0.563	0.354	0.563	0.563	0.528	0.354	0.563	0.771
20243	18	890721	931992	41271	0.097	0.394	0.537	0.481	0.537	0.366	0.537	0.537	0.472	0.394	0.514	0.681

Table 2. Characteristics of the 24 *S. bovis*-like genomic tracts detected in at least one *S. haematobium* isolate and Dxy values computed between each *S. haematobium* isolate and all the *S. bovis* isolates within each of these 24 *S. bovis*-like genomic tracts. The scaffold 020018 in bold and framed in the table contains the *S. bovis*-like *Invadolysin* gene (Fig 4). Red coloration levels reflect small (dark red) to high (light red) Dxy values. Extreme low Dxy values (i.e. <0.102) are in bold.

https://doi.org/10.1371/journal.ppat.1009313.t002

from the scaffold 020018 contains, among other genes, the *Invadolysin* gene. Finally, only three genes were found within the five *S. haematobium*-type genomic tracts that were found in at least one of the *S. bovis* isolate (S2 Table).

Discussion

All S. *haematobium* isolates and the natural Corsican hybrid form a unique genomic cluster

Based on our clustering approach, we found that the hybrid isolate from Corsica, together with all nominal *S. haematobium* isolates, group into a unique cluster harboring low genomic diversity compared to the *S. bovis* cluster (Fig 2). This result was somewhat unexpected given that the Corsican isolate is classified as a *S. haematobium-bovis* hybrid based on discordant mt / nuclear genetic markers (*S.bovis*-type *CO1* mitochondrial haplotype and *S. haematobium*-type *ITS* nuclear haplotype). Moreover, we have previously shown the Corsican isolate's entire mitochondria was of *S. bovis* origin, with an estimated 23% of its nuclear genome of *S. bovis* origin, whereas ~77% of *S. haematobium* origin, proving that this was not a first generation hybrid but that there had been introgression from *S. bovis* into *S. haematobium* [21]. Hence, we might have expected that this sample would have an intermediate location between the *S. haematobium* and the *S. bovis* clusters in the PCA space although closer to the *S. haematobium* cluster. Two non-exclusive main hypotheses might explain these results.

First, on a methodological perspective, our approach was based on the development of SNPs from supposedly orthologous regions between these two sister species. In particular, the initial sequencing reads obtained from each isolate were aligned to the S. haematobium reference genome (Egyptian strain) [41] and only those that aligned properly were considered for subsequent analyses. Kincaid-Smith et al. [21] assessed the introgression level of the Corsican strain using an independent analytical approach based on the alignment of sequencing reads on a concatenated genome combining the genomes of both parental species (i.e. S. haematobium and S. bovis). Sequences were next aligned according to the best mapping location over the two genomes and the proportion of sequences that aligned to the S. bovis and to the S. haematobium regions of the concatenated genome were then used to assess its introgression level. Using our approach, we might have missed some S. bovis introgressed genomic (private) regions in the Corsican isolate that did not align properly on the S. haematobium reference genome. Such private regions are expected to be rare due to the high genomic similarity between these two species [24]. Nevertheless, and if so, we might expect that using a S. bovis instead of a S. haematobium reference genome for the initial alignment of sequencing reads and for the development of SNPs would allow detection of such private S. bovis genomic regions and hence tease apart the Corsican isolate from the S. haematobium group. However, an additional PCA analysis conducted on SNPs developed from the alignment of the sequencing reads against the S. bovis reference genome, strictly led to the same conclusions (S2 Fig).

The second hypothesis is that all isolates originally described as *S. haematobium* are in fact introgressed, to some extent, with *S. bovis* similarly as with the hybrid isolate from Corsica. In this regard, two recent studies strongly suggest that at least two lineages initially identified as *S. haematobium* (based on morphological, ecological and genetic diagnostics), including the laboratory strain from which the reference genome was assembled (Egyptian strain) [41], were in fact introgressed with *S. bovis* although moderately (i.e. 3–8%) [23,24]. Based on these previous studies and because of the results obtained in the present study, we thus investigated the presence of genomic tracts from *S. bovis* that could have been acquired in the genomes of the *S. haematobium* isolates. We also investigated the presence of genomic tracts from *S. haematobium* within all the *S.bovis* isolates that could result from past introgression events between these two sister species.

Evidence for past introgression events between S. bovis and S. haematobium

In total, 24 genomic tracts with extremely low levels of divergence between at least one of the *S. haematobium* isolate and all of the *S. bovis* isolates were identified (Table 2; Fig 3). These genomic tracts could result from their transfer from *S. bovis* to at least some *S. haematobium* lineages during past introgression events. Importantly however, a decrease in divergence between the two sister species is also expected locally at some genomic positions due to shared ancestral polymorphism even in the absence of introgression [2,43]. Admittedly, our current analyses do not account for such ancestral polymorphism. However, several results discredit this hypothesis in favor of a past introgression scenario.

First, the identified genomic tracts characterised by extremely low Dxy values were sometimes of considerable size (up to 630.020 bp) which is not expected under the retention of ancestral polymorphism scenario [44]. Second, contrary to what we found among *S. haematobium* isolates, very few genomic regions with significantly decreased divergence between each of the *S. bovis* isolates and all of the *S. haematobium* isolates were detected (N = 5). Under an ancestral polymorphism scenario we might have expected detecting reciprocal patterns of genomic similarities between each *S. bovis* isolate and *S. haematobium*. In this regard, our results thus not only support an introgression scenario between these two sister species but





https://doi.org/10.1371/journal.ppat.1009313.g003

also that introgression was most likely unidirectional and has led to the transfer of some admixted tracts from *S. bovis* into at least some of the *S. haematobium* isolates. However, this latter conclusion needs to be taken with caution since only six *S. bovis* isolates were included in this study and we cannot rule out the existence of some *S. bovis* lineages introgressed with *S. haematobium* across Africa. Third, among the 24 *S. bovis*-like genomic tracts found over the *S. haematobium* isolates, one contains the *Invadolysin* gene (Table 2; Fig 4). This gene was

Table 3. Characteristics of the five *S. haematobium*-like genomic tracts detected in at least one *S. bovis* isolate and Dxy values computed between each *S. bovis* isolate and all the *S. haematobium* isolates within each of these five *S. haematobium*-like genomic tracts. Red coloration levels reflect small (in dark red) to high (in light red) Dxy values. Extreme low Dxy values (i.e. <0.102) are in bold.

Scaffold identity	Number of SNPs	Start position	End position	Tract size (bp)	Ethiopia	Kenya	Nigeria	Senegal	Spain	Uganda
20100	9	10330900	10379570	48670	0.313	0.096	0.591	0.045	0.48	0.146
20112	8	345456	347912	2456	0.568	0.068	0.506	0.443	0.631	0.256
20168	8	605485	645215	39730	0.193	0.506	0.131	0.688	0.006	0.756
20173	8	6220527	6240433	19906	0.341	0.301	0.403	0.091	0.182	0.301
20244	13	64302	78634	14332	0.105	0.259	0.066	0.259	0.643	0.182

https://doi.org/10.1371/journal.ppat.1009313.t003



Fig 4. Focus on a *S. bovis*-like genomic tract detected in some *S. haematobium* isolates along the genomic scaffold 020018. A. Dxy values computed between each of the 12*S. haematobium* isolates and the experimental F1 hybrid and all of the *S. bovis* isolates along the genomic scaffold 020018. Extremely low values of Dxy were found between three *S. haematobium* isolates (i.e. Corsica, Mali and Egypt) and *S. bovis* according to the Dxy = 0.102 fixed threshold (see material and methods). Dxy values corresponding to the isolates from Corsica, Mali and Egypt are in blue, red and yellow respectively. Dxy values computed between the experimental F1 hybrid and all *S.bovis* isolates are represented by a dashed black line. Dxy values computed between all other isolates and *S. bovis* isolates are in grey. **B.** Representation of the allelic state of each isolate at each SNP along the 212.351 bp *S.bovis*-like genomic tract detected in the isolates from Corsica, Mali and Egypt that contain, among other genes, the *Invadolysin* gene previously identified in a *S. haematobium* population from Niger and in a *S. haematobium* reference strain (Egyptian strain) [41]. SNPs at which isolates are homozygous and harboring an allele identical or different to the genome of reference (Egyptian strain) [41] are in black and orange respectively. SNPs at which isolates are heterozygous are in grey.

https://doi.org/10.1371/journal.ppat.1009313.g004

previously identified to be of *S. bovis*-type in some introgressed individuals of a *S. haematobium* population in Niger and also in the isolate which is used as the *S. haematobium* reference [23]. In the present study, this same *S. bovis*-type sequence of the *Invadolysin* gene was found in the *S. haematobium* isolates from Egypt, Corsica and Mali. Moreover, *S. haematobium* isolates from Sudan, Cameroon and Guinea-Bissau also displayed very low Dxy values at this genomic tract although slightly higher than the arbitrarily fixed threshold (0.117, 0.145 and 0.173 respectively). Together, these results suggest that all of these isolates likely originate from the same ancient introgression event at the origin of the introgressed lineage from Niger identified previously that occurred ~108–613 generations ago according to Platt et al. [23].

Interestingly, the number of admixed genomic tracts greatly varies among the *S. haemato-bium* isolates. Isolates from Corsica and Mali respectively displayed 9 and 10 admixture tracts while all other isolates displayed up to 4 and the isolate from Madagascar none (Table 2). This

result is particularly interesting since Madagascar is the only country sampled in this study in which *S. haematobium* lives in allopatry (i.e. where *S. bovis* was never detected). Accordingly, we argue that the isolate from Madagascar analyzed in the present study is the unique representative of what could be considered a pure *S. haematobium*. Zanzibar is another geographic region where *S. bovis* was considered absent until its first detection in 2016 [45]. Hence, we might have expected not to detect traces of potential introgression from *S. bovis* within this isolate. However, we did find a potential *S. bovis* admixture tract in the isolate from Zanzibar, which was also found in the isolates from Senegal and Guinea-Bissau. This result thus suggests that the *S. haematobium* from Zanzibar isolate analyzed in this study results from an old introgression event between *S. haematobium* and *S. bovis* and that the evolutionary history of this isolate is thus independent to the one from Madagascar. Importantly however, this unique potential admixture tract identified in the isolate from Zanzibar is relatively small (i.e. 26.593 bp) and contain few SNPs (N = 13; Table 2). Therefore, we cannot formerly rule out the possibility that this *S. bovis*-like genomic tract truly reflects a shared ancestral polymorphism rather than an introgression event between the two species.

Whether all S. haematobium isolates that have been identified as introgressed (i.e. all but the isolate from Madagascar) originate from a single common past introgression event between S. bovis and S. haematobium remains an unresolved question. On one hand, the fact that three isolates (six if considering Dxy values higher than the 0.102 threshold) among the 12 S. haematobium isolates studied displayed the same genomic tract that includes the S. bovislike Invadolysin gene strongly supports a common introgression event at the origin of these isolates and the previously identified introgressed within S. haematobium from Niger [23]. On the other hand, most of the S. bovis-like genomic tracts identified within the isolates are generally specific to a single isolate only (Table 2). These later results suggest either a common introgression event that occurred a long time ago, which is in line with the single introgression event identified by Platt et al. [23] that has been dated to ~108-613 generations ago. Under this scenario, different genomic tracts including the one that contains the *Invadolysin* gene could have been fixed in some populations while lost in other populations through genetic drift. Alternatively, the observed heterogeneous pattern of admixed tracts from S. bovis within the S. haematobium isolates could reflect several independent past introgression events. Because the identified admixture tract, although generally longer than that expected under an ancestral polymorphism scenario, are relatively small in most S. haematobium isolates, the potential independent past introgression events are also likely to be ancient. Indeed, it is expected that successive recombination events over generations since the initial introgression event can lead to a drastic size reduction of admixture tracts in the resulting introgressed lineage [46].

On a final note, it is likely that the degree of genomic introgression from *S. bovis* to the *S. haematobium* isolates observed in this study is underestimated. In fact, only 53% of the overall genome was screened for potential genomic regions with a decrease of genomic divergence between these sister species in particular because we applied stringent filters applied for developing SNPs (i.e. no missing data) and we mainly focused on genomic regions with initial genomic variation (i.e. were SNPs were present in at least one isolate). This approach is likely to provide a conservative vision of introgression patterns between species.

Implications for the evolution of Schistosoma haematobium

Our results are in line with the recent hypothesis stating that at least one introgression event occurred a long time ago between *S. haematobium* and *S. bovis* and resulted in the introgression of some *S. bovis* genomic tracts into several *S. haematobium* lineages [23]. Moreover, our

study suggests that introgression resulting from past interspecific mating is observed in all of the analysed S. haematobium lineages except the allopatric lineage originating from Madagascar. Whether these isolates originate from a single or different past introgression event(s) still remains to be investigated. In the same vein, whether interspecific interactions between S. haematobium and S. bovis are still ongoing remains an open question. So far, accumulating evidence indicates that hybridisation events between S. haematobium and S. bovis are scarce even in regions where these two species live in sympatry (e.g. Senegal) [47]. We also found that all but the isolate from Madagascar in fact exhibited S. bovis-like genomic tracts within their genomes sometimes longer and in greater number than those found in the isolate from Corsica (e.g. the isolate from Mali). This result indicates that past introgression events between S. haematobium and S. bovis have not only influenced the genomic background of the parasites considered as hybrids so far but certainly concern most S. haematobium lineages across Africa. On a biological point of view, these two species use different definitive hosts, the former being associated with humans and the latter livestock. Within their respective hosts, they also differ in their tropisms (S. haematobium being urogenital and S. bovis being mesenteric). These two specific biological characteristics greatly limit possible interspecific reproduction events. Importantly however, individuals harboring both a nuclear and a mitochondrial genetic signature of S. bovis are seldom detected in humans [19,20]. More recently, individuals biologically characterized as S. haematobium (according to egg morphology and chronobiology) and genetically identified as S. haematobium-bovis hybrids were detected in cattle for the first time in Benin [48]. Together these results suggest that current simultaneous infection of animals by S. haematobium and of humans by S. bovis may occur at some level but it is not common. However, the genomic composition of those individuals from Benin has not been characterized and the possibility that these individuals in fact belong to introgressed lineages cannot be ruled out. We believe that the set of SNPs developed in this study could shed light on the genomic nature of atypical schistosome samples and provide new insights on how specific biological and genetics features are conserved within each species. The genomic pattern of the S. haematobium-bovis experimentally generated F1 hybrid observed in the current study indicates that this will be a relevant tool to detect first and early generation hybrids in natural populations. It will be of interest to combine these methods with observations on mito-nuclear profiles to clarify the extent of ongoing hybridization, changes in host specificity and potential zoonotic transmission.

Additionally, we found that *S. haematobium* and/or hybrid lineages also harbored different introgressed *S. bovis* genomic regions. In this respect, it is very likely that important diversity exists in terms of introgressed genomic regions among *S. haematobium* lineages in Africa and that such introgressed genes could have been maintained independently as a result of different selective pressures in different epidemiological contexts. Such diverse introgression patterns could explain some variations in virulence and compatibility with snails in *S. haematobium* natural populations [49,50]. We advocate that more genomic analyses are thus required to quantify such heterogeneity in introgression patterns and validate the hypothetical link between introgression patterns and these important life-history traits.

On a final note, our results also clearly indicate that the *S. haematobium* cluster harbors very little genetic diversity compared to *S. bovis* and no geographic structure was detected in *S. haematobium* at the African continent scale (Fig 2). These general patterns across Africa are in line with previous results obtained at a microgeographic scale based on both mitochondrial and microsatellite markers [47,51]. More generally, these results confirm, at global geographical and genomic scales, that *S. haematobium* is a genetically depauperate species compared to other *Schistosoma* species [51–53]. Webster et al. [52] suggested that such low genetic diversity most likely reflects a possible recent (re-)invasion of *S. haematobium* in Africa from a small

number of founding individuals from the Arabian Peninsula and a rapid expansion from East to West. The lack of spatial structuring pattern of genetic differentiation observed in *S. haema-tobium* across Africa is in line with this hypothesis. Under such a scenario, one could emit the hypothesis that introgression events between *S. bovis* and *S. haematobium* arose before the recolonisation of *S. haematobium* across Africa.

Conclusion

Our study reveals that all the *S. haematobium* isolates analyzed here except the one from Madagascar, to some extent, harbor signals of introgression from *S. bovis* and constitute a cluster with very low genetic diversity at the African continent scale. Moreover, *S. haematobium* lineages differ from one another by the conserved introgressed genomic regions originating from *S. bovis* due either to genomic drift or local adaptive processes. This means that important variation might exist in *S. haematobium* populations concerning the genomic regions acquired through introgression and maintained over generations with possible different evolutionary outcomes in terms of interactions with their hosts and epidemiological dynamics. We argue that the SNPs developed in this study will constitute a toolbox for future studies aiming at better understanding the evolutionary history of the two sister species *S. haematobium* and *S. bovis* and *S. haematobium-bovis* hybrids. This approach will allow higher resolution of the genetic makeup of natural schistosome populations and will facilitate deeper insights into the epidemiology of schistosomiasis across Africa.

Supporting information

S1 Table. Sequencing and alignment statistics obtained for each isolate analysed in this study.

(XLSX)

S2 Table. Details of the 29 genomic tracts that display a significant decrease in genomic divergence between either at least one *S. haematobium* isolate and all *S. bovis* isolates or at least one *S. bovis* isolate and all *S. haematobium* isolates. Annotated genes that were identified in these genomic tracts are also listed. (XLSX)

S1 Fig. Plots obtained from a PCA analysis based on 56.141 SNPs developed from the initial alignment of sequencing reads on the *S. haematobium* genome of reference. Isolates are projected within two-dimensional spaces defined from A) axes 1 and 2; B) axes 2 and 3; C) axes 2 and 4 and D) axes 3 and 4. Isolates identified as *S. bovis* are in red and the F1 laboratory hybrid and isolates identified as *S. haematobium* are in grey. (PDF)

S2 Fig. Plots obtained from a PCA analysis based on 172.046 SNPs developed from the initial alignment of sequencing reads on the *S. bovis* genome of reference. Isolates are projected within two-dimensional spaces defined from A) axes 1 and 2; B) axes 2 and 3; C) axes 2 and 4 and D) axes 3 and 4. Isolates identified as *S. bovis* are in red and the F1 laboratory hybrid and isolates identified as *S. haematobium* are in grey. (PDF)

Acknowledgments

We would like to warmly thank all the people that have helped during the sample collection including people from G. Riveau's team 'Espoir pour la Santé' in Senegal, Felicité Flore

Djuikwo-Teukeng from Cameroon, Etienne Angora from Ivory Coast and Kalkidan Mekete from Ethiopia. We thank Dr Aidan Emery (Principal Investigator of SCAN, Schistosomiasis Collection at the Natural History Museum) for his support for the project. We are grateful to Muriel Rabone (SCAN, NHM) for assistance with sample databasing and retrieval. Material provided via SCAN was originally collected between 1985 and 2006 and we acknowledge the support and generosity of our collaborators and partners involved in the collections, particularly in relation to the endemic countries involved, and past colleagues that maintained the live material. We would also like to thank the Biomedical Research Center for kindly providing the *S. haematobium* sample from the Egyptian stain. Finally, the authors warmly thank the Bio-Environnement platform (University of Perpignan) for sequencing and bioinformatics services. This study is set within the framework of the "Laboratoire d'Excellence (LABEX)" TULIP (ANR-10-LABX-41).

Author Contributions

Conceptualization: Olivier Rey.

Data curation: Olivier Rey.

Formal analysis: Olivier Rey, Eve Toulza, Cristian Chaparro, Jean-François Allienne, Eglantine Mathieu-Begné.

Funding acquisition: Olivier Rey, Jérôme Boissier.

Investigation: Olivier Rey.

Methodology: Olivier Rey, Eve Toulza, Cristian Chaparro, Jean-François Allienne.

Project administration: Olivier Rey.

Resources: Jean-François Allienne, Fiona Allan, David Rollinson, Bonnie L. Webster.

Writing - original draft: Olivier Rey.

Writing – review & editing: Olivier Rey, Eve Toulza, Cristian Chaparro, Julien Kincaid-Smith, Eglantine Mathieu-Begné, Fiona Allan, David Rollinson, Bonnie L. Webster, Jérôme Boissier.

References

- Mallet J. Hybridization as an invasion of the genome. Trends Ecol Evol. 2005; 20:229–37. https://doi. org/10.1016/j.tree.2005.02.010 PMID: 16701374
- Baack EJ. Rieseberg LH. A genomic view of introgression and hybrid speciation. Curr Opin Genet Dev. 2007; 17:513–8. https://doi.org/10.1016/j.gde.2007.09.001 PMID: 17933508
- 3. Runemark A, Vallejo-Marin M, Meier JI. Eukaryote hybrid genomes. PLoS Genet. 2019; 15:e1008404. https://doi.org/10.1371/journal.pgen.1008404 PMID: 31774811
- Chen N, Cai Y, Chen Q, Li R, Wang K, Huang Y, et al. Whole-genome resequencing reveals world-wide ancestry and adaptive introgression events of domesticated cattle in East Asia. Nat Commun. 2018; 9:2337. https://doi.org/10.1038/s41467-018-04737-0 PMID: 29904051
- Mitchell N, Owens GL, Hovick SM, Rieseberg LH, Whitney KD. Hybridization speeds adaptive evolution in an eight-year field experiment. Sci Rep. 2019; 9:6746. https://doi.org/10.1038/s41598-019-43119-4 PMID: 31043692
- Racimo F, Marnetto D, Huerta-Sánchez E. Signatures of Archaic Adaptive Introgression in Present-Day Human Populations. Mol Biol Evol. 2016; msw216. <u>https://doi.org/10.1093/molbev/msw216</u> PMID: 27756828
- Barbato M, Hailer F. Orozco-terWengel P, Kijas J, Mereu P, Cabras P, et al. Genomic signatures of adaptive introgression from European mouflon into domestic sheep. Sci Rep. 2017; 7:7623. <u>https://doi.org/10.1038/s41598-017-07382-7</u> PMID: 28790322

- Hedrick PW. Adaptive introgression in animals: examples and comparison to new mutation and standing variation as sources of adaptive variation. Mol Ecol. 2013; 22:4606–18. <u>https://doi.org/10.1111/mec.</u> 12415 PMID: 23906376
- King KC, Stelkens RB, Webster JP, Smith DF, Brockhurst MA. Hybridization in Parasites: Consequences for Adaptive Evolution, Pathogenesis, and Public Health in a Changing World. Rall GF, editor. PLoS Pathog 2015; 11: e1005098. https://doi.org/10.1371/journal.ppat.1005098 PMID: 26336070
- Brasier CM, Cooke DEL, Duncan JM. Origin of a new Phytophthora pathogen through interspecific hybridization. Proc Natl Acad Sci. 1999; 96:5878–83. https://doi.org/10.1073/pnas.96.10.5878 PMID: 10318978
- 11. Arnold ML. Natural hybridization and the evolution of domesticated, pest and disease organisms. Mol Ecol. 2004; 13:997–1007. https://doi.org/10.1111/j.1365-294X.2004.02145.x PMID: 15078439
- Detwiler JT, Criscione CD. An Infectious Topic in Reticulate Evolution: Introgression and Hybridization in Animal Parasites. Genes. 2010; 1:102–23. https://doi.org/10.3390/genes1010102 PMID: 24710013
- Rollinson D. The genus Schistosoma: a taxonomic appraisal. The biology of schistosomes from genes to latrines. 1987.
- Jones BP, Norman BF, Borrett HE, Attwood SW, Mondal MMH, Walker AJ, et al. Divergence across mitochondrial genomes of sympatric members of the *Schistosoma indicum* group and clues into the evolution of *Schistosoma spindale*. Sci Rep. 2020; 10:2480. <u>https://doi.org/10.1038/s41598-020-57736-x PMID: 32051431</u>
- Leger E, Webster JP. Hybridizations within the Genus Schistosoma: implications for evolution, epidemiology and control. Parasitology. 2016:1–16. <u>https://doi.org/10.1017/S0031182016001190</u> PMID: 27572906
- Webster BL, Tchuenté LT, Jourdane J, Southgate VR. The interaction of *Schistosoma haematobium* and *S. guineensis* in Cameroon. J Helminthol. 2005; 79:193–7. <u>https://doi.org/10.1079/joh2005306</u> PMID: 16153312
- Huyse T, Van den Broeck F, Hellemans B, Volckaert FAM, Polman K. Hybridisation between the two major African schistosome species of humans. Int J Parasitol. 2013; 43:687–9. <u>https://doi.org/10.1016/j.ijpara.2013.04.001</u> PMID: 23643461
- Webster BL, Diaw OT, Seye MM, Webster JP, Rollinson D. Introgressive hybridization of *Schistosoma haematobium* group species in Senegal: Species barrier break down between ruminant and human schistosomes. PLoS Negl Trop Dis. 2013; 7:e2110. <u>https://doi.org/10.1371/journal.pntd.0002110</u> PMID: 23593513
- Boissier J, Grech-Angelini S, Webster BL, Allienne J-F, Huyse T, Mas-Coma S, et al. Outbreak of urogenital schistosomiasis in Corsica (France): an epidemiological case study. Lancet Infect Dis. 2016; 16:971–9. https://doi.org/10.1016/S1473-3099(16)00175-4 PMID: 27197551
- Angora EK, Allienne J-F, Rey O, Menan H, Touré AO, Coulibaly JT, et al. High prevalence of *Schistosoma haematobium* × *Schistosoma bovis* hybrids in schoolchildren in Côte d'Ivoire. Parasitology. 2019:1–8. https://doi.org/10.1017/S0031182019001549 PMID: 31727202
- Kincaid-Smith J, Tracey A, de Carvalo Augusto R, Bulla I, Holroyd N, Rognon A, et al. Whole genome sequencing and morphological analysis of the human-infecting schistosome emerging in Europe reveals a complex admixture between *Schistosoma haematobium* and *Schistosoma bovis* parasites. BioRxiv. 2018; 387969.
- Kincaid-Smith J. Modification des traits d'histoire de vie au cours de l'hybridation et analyse des mécanismes moléculaires sous-jacents chez les parasites plathelminthes du genre Schistosoma. PhD. 2018.
- Platt RN, McDew-White M, Le Clec'h W, Chevalier FD, Allan F, Emery AM, et al. Ancient Hybridization and Adaptive Introgression of an Invadolysin Gene in Schistosome Parasites. Shapiro B, editor. Mol Biol Evol 2019; 36: 2127–2142. https://doi.org/10.1093/molbev/msz154 PMID: 31251352
- Oey H, Zakrzewski M, Gravermann K, Young ND, Korhonen PK, Gobert GN, et al. Whole-genome sequence of the bovine blood fluke *Schistosoma bovis* supports interspecific hybridization with *S. haematobium*. Platt RN, editor. PLoS Pathog 2019; 15: e1007513. <u>https://doi.org/10.1371/journal.ppat.</u> 1007513 PMID: 30673782
- Emery AM, Allan FE, Rabone ME, Rollinson D. Schistosomiasis collection at NHM (SCAN). Parasit Vectors. 2012; 5:185. https://doi.org/10.1186/1756-3305-5-185 PMID: 22943137
- Barber KE, Mkoji GM, Loker ES. PCR-RFLP analysis of the ITS2 region to identify *Schistosoma haema-tobium* and *S. bovis* from Kenya. The American journal of tropical medicine and hygiene. 2000; 62:434–40. https://doi.org/10.4269/ajtmh.2000.62.434 PMID: 11220757
- 27. Lockyer AE, Olson PD, Østergaard P, Rollinson D, Johnston DA, Attwood SW, et al. The phylogeny of the Schistosomatidae based on three genes with emphasis on the interrelationships of *Schistosoma*

Weinland, 1858. Parasitology. 2003; 126:203–24. https://doi.org/10.1017/s0031182002002792 PMID: 12666879

- Afgan E, Baker D, Batut B, Van Den Beek M, Bouvier D, Čech M, et al. The Galaxy platform for accessible, reproducible and collaborative biomedical analyses: 2018 update. Nucleic Acids Res. 2018; 46: W537–44. https://doi.org/10.1093/nar/gky379 PMID: 29790989
- **29.** Andrews S. FastQC: a quality control tool for high throughput sequence data. Babraham Bioinformatics, Babraham Institute, Cambridge, United Kingdom; 2010.
- Langmead B, Salzberg SL. Fast gapped-read alignment with Bowtie 2. Nat Methods. 2012; 9:357–9. https://doi.org/10.1038/nmeth.1923 PMID: 22388286
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics. 2009; 25:2078. <u>https://doi.org/10.1093/bioinformatics/btp352</u> PMID: 19505943
- DePristo MA, Banks E, Poplin R, Garimella KV, Maguire JR, Hartl C, et al. A framework for variation discovery and genotyping using next-generation DNA sequencing data. Nat Genet. 2011; 43:491–8. https://doi.org/10.1038/ng.806 PMID: 21478889
- Danecek P, Auton A, Abecasis G, Albers CA, Banks E, DePristo MA, et al. The variant call format and VCFtools. Bioinformatics. 2011; 27:2156–8. <u>https://doi.org/10.1093/bioinformatics/btr330</u> PMID: 21653522
- Thioulouse J, Dray S. Interactive multivariate data analysis in R with the ade4 and ade4TkGUI packages. J Stat Softw. 2007; 22:1–14.
- **35.** Core Team R. R: a language and environment for statistical computing. R Foundation for statistical computing, Vienna. 2013.
- Kassambara A, Mundt F. Package 'factoextra.'. Extract and visualize the results of multivariate data analyses. 2017:76.
- Pfeifer B, Wittelsbürger U, Ramos-Onsins SE, PopGenome LMJ. An Efficient Swiss Army Knife for Population Genomic Analyses in R. Mol Biol Evol. 2014; 31:1929–36. https://doi.org/10.1093/molbev/ msu136 PMID: 24739305
- 38. Nei M. Genetic distance between populations. Am Nat. 1972; 106:283-92.
- Gruber B, Unmack PJ, Berry OF. Georges A. dartr: An r package to facilitate analysis of SNP data generated from reduced representation genome sequencing. Mol Ecol Resour. 2018; 18:691–9. <u>https://doi.org/10.1111/1755-0998.12745 PMID: 29266847</u>
- **40.** Paradis E. Schliep K. ape 5.0: an environment for modern phylogenetics and evolutionary analyses in R. Bioinformatics. 2019; 35:526–8. https://doi.org/10.1093/bioinformatics/bty633 PMID: 30016406
- Stroehlein AJ, Korhonen PK, Chong TM, Lim YL, Chan KG, Webster B, et al. High-quality Schistosoma haematobium genome achieved by single-molecule and long-range sequencing. GigaScience. 2019; 8: giz108. https://doi.org/10.1093/gigascience/giz108 PMID: 31494670
- Robinson JT, Thorvaldsdóttir H, Winckler W, Guttman M, Lander ES, Getz G, et al. Integrative Genomics Viewer. Nat Biotechnol. 2011; 29:24–6. https://doi.org/10.1038/nbt.1754 PMID: 21221095
- Twyford AD, Ennos RA. Next-generation hybridization and introgression. Heredity. 2012; 108:179–89. https://doi.org/10.1038/hdy.2011.68 PMID: 21897439
- Racimo F, Sankararaman S, Nielsen R, Huerta-Sánchez E. Evidence for archaic adaptive introgression in humans. Nat Rev Genet. 2015; 16:359–71. https://doi.org/10.1038/nrg3936 PMID: 25963373
- Pennance T, Ame SM, Amour AK, Suleiman KR, Allan F, Rollinson D, et al. Occurrence of *Schistosoma bovis* on Pemba Island, Zanzibar: implications for urogenital schistosomiasis transmission monitoring. Parasitology. 2018; 145:1727–31. https://doi.org/10.1017/S0031182018001154 PMID: 30086805
- 46. Turissini DA, Matute DR. Fine scale mapping of genomic introgressions within the *Drosophila yakuba* clade. Begun DJ, editor PLoS Genet. 2017; 13:e1006971. https://doi.org/10.1371/journal.pgen. 1006971 PMID: 28873409
- Boon NAM, Mbow M, Paredis L, Moris P, Sy I, Maes T, et al. No barrier breakdown between human and cattle schistosome species in the Senegal River Basin in the face of hybridisation. Int J Parasitol. 2019; 49:1039–48. https://doi.org/10.1016/j.ijpara.2019.08.004 PMID: 31734338
- Savassi BAES, Mouahid G, Lasica C, Mahaman S-DK, Garcia A, Courtin D, et al. Cattle as natural host for *Schistosoma haematobium* (Bilharz, 1852) Weinland, 1858 x *Schistosoma bovis* Sonsino, 1876 interactions, with new cercarial emergence and genetic patterns. Parasitol Res. 2020 [cited 3 Jun 2020]. https://doi.org/10.1007/s00436-020-06709-0 PMID: 32468189
- 49. Rollinson D, Stothard JR, Southgate VR. Interactions between intermediate snail hosts of the genus Bulinus and schistosomes of the Schistosoma haematobium group. Parasitology. 2001; 123:245–60. https://doi.org/10.1017/s0031182001008046 PMID: 11769287

- Afifi M, Jiman-Fatani A, Al-Hussainy N, Al-Rabia M, Bogari A. Genetic diversity among natural populations of *Schistosoma haematobium* might contribute to inconsistent virulence and diverse clinical outcomes. J Microsc Ultrastruct. 2016; 4:222. https://doi.org/10.1016/j.jmau.2016.04.002 PMID: 30023230
- Djuikwo-Teukeng FF, Kouam Simo A, Allienne J-F, Rey O, Njayou Ngapagna A, Tchuem-Tchuente LA, et al. Population genetic structure of *Schistosoma bovis* in Cameroon. Parasit Vectors. 2019; 12 (56). https://doi.org/10.1186/s13071-019-3307-0 PMID: 30678712
- Webster BL, Emery AM, Webster JP, Gouvras A, Garba A, Diaw O, et al. Genetic Diversity within Schistosoma haematobium: DNA Barcoding Reveals Two Distinct Groups. Jex AR, editor. PLoS Negl Trop Dis. 2012; e1882:6. https://doi.org/10.1371/journal.pntd.0001882 PMID: 23145200
- 53. Gower CM, Gouvras AN, Lamberton PHL, Deol A, Shrivastava J, Mutombo PN, et al. Population genetic structure of *Schistosoma mansoni* and *Schistosoma haematobium* from across six sub-Saharan African countries: Implications for epidemiology, evolution and control. Tropica. 2013; 128:261–74. https://doi.org/10.1016/j.actatropica.2012.09.014 PMID: 23041540
- Silva ML, Vicente FS, Avelino IC, Martin VR. Susceptibility of *Planorbarius metidjensis* from Portugal and Spain to *Schistosoma bovis* from Salamanca, Spain. Malacologia. 1977; 16:251–4. PMID: 904368
- 55. Lewis FA, Yung-san Liang NR, Knight M. The NIH-NIAID schistosomiasis resource center. PLoS Negl Trop Dis. 2008; 2. https://doi.org/10.1371/journal.pntd.0000267 PMID: 18665228